

vided by Repositorio Institucional de la Biblioteca del Banco de España

2018

THE EFFECTS OF TAX CHANGES ON ECONOMIC ACTIVITY: A NARRATIVE APPROACH TO FREQUENT ANTICIPATIONS

Sandra García-Uribe

Documentos de Trabajo N.º 1828

BANCODEESPAÑA

Eurosistema

THE EFFECTS OF TAX CHANGES ON ECONOMIC ACTIVITY: A NARRATIVE APPROACH TO FREQUENT ANTICIPATIONS

THE EFFECTS OF TAX CHANGES ON ECONOMIC ACTIVITY: A NARRATIVE APPROACH TO FREQUENT ANTICIPATIONS ^(†)

Sandra García-Uribe (**)

BANCO DE ESPAÑA

Documentos de Trabajo. N.º 1828 2018

^(*) I am very grateful to Manuel Arellano for his guidance and encouragement. I am also very grateful to Dante Amengual, Samuel Bentolila, Guillermo Caruana, Diego García, Gerard Llobet, Mónica Martínez-Bravo, Pedro Mira, Diego Puga and Enrique Sentana for their many comments and suggestions, as well as to seminar participants at CEMFI and Banco de España. I gratefully acknowledge the Spanish Ministry of Economy and Competition FPI Grant BES-2014-070531 for financial support. Errors and omissions are exclusively my own. (**) Banco de España. Alcalá, 48, 28014 Madrid, Spain. Email: sandra.garcia.uribe@bde.es. Personal website: http://sites.google.com/view/garciauribe.

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2018

ISSN: 1579-8666 (on line)

Abstract

This paper studies the effects of anticipations of tax changes in the USA through the release of tax news in the media. I construct a new measure that captures the anticipation of tax bill approvals by exploiting the content of news in the US television. Since this information typically flows faster than standard measures of GDP, I propose a mixed frequency dynamic factor model to estimate both the economic activity latent factor and the effects of anticipated tax shocks on it. I find that onemonth-ahead media anticipations of tax approvals significantly stimulate current economic activity. This stimulation comes from anticipations of tax cuts.

Keywords: fiscal policy, taxation, mass media, information, beliefs, random forests.

JEL classification: E62, H20, N12, D80.

Resumen

Este artículo estudia los efectos que anticipaciones de cambios impositivos a través de noticias en los medios pudieron tener en la actividad económica de Estados Unidos. Para ello construyo una nueva medida que contiene el nivel de anticipación de la posible aprobación de proyectos de ley fiscal en el Congreso americano explotando el contenido de las noticias que se publicaron en la televisión norteamericana. Dado que esta información fluye más rápido que las medidas estándar de PIB, propongo un modelo dinámico de frecuencias mixtas para estimar tanto el factor inobservable de actividad económica como los efectos de los potencialmente previsibles cambios fiscales sobre él. Encuentro que anticipaciones de cambios en la presión fiscal a un mes vista con la información de los medios estimulan significativamente la actividad económica, siendo las anticipaciones de reducciones en la presión fiscal las que determinan el signo de este resultado.

Palabras clave: política fiscal, impuestos, medios de comunicación, información, creencias, bosques aleatorios.

Códigos JEL: E62, H20, N12, D80.

1 Introduction

Prior to the approval of laws, there is often widespread information about the progress of bills. This information may be valuable for the forecasts that agents make about the economic environment in the future. In the context of fiscal policy, agents may be very attentive to information on these bills if by doing so they are able to anticipate changes that will affect them. This paper provides a way to account for the economic responses to anticipation of tax shocks when there is still uncertainty about the approval of tax bills.

Mass media are a central player in the transmission of public information to society. As reporters and analysts of fiscal bills during their process of elaboration and until approval, they provide information that may be worth to the public's forecast of the success of potential new fiscal policy. This is an anticipation channel that has not been considered before in the literature but one that as this paper shows is relevant to economic activity.

In this paper I introduce a new measure of mass media anticipations of tax bill approvals by exploiting the content of news in the US television during the period 1968-2007. My measure of mass media anticipations is formulated as the predicted probability of a tax bill approval at future periods conditional on the available information in mass media news at the current period. To address this quantification challenge I use a machine learning algorithm from the family of Classification and Regression Trees (CART). CART methods allow the researcher to consider large number of variables for a classification task, whereas standard discrete choice models such as Probit or Logit cannot handle this problem. They are highly non-linear models which help learning the features that better fit the data in a very flexible way. Nonetheless, they suffer from overfitting when the data is very noisy and of bias when there is a high correlation among the features. The news data suffers from all these issues. To overcome them I make use of automatic text analysis techniques to reduce the noise in the data results.

Time aggregation in the analysis of quarterly economic indicators may mask important anticipation effects, since information through the news may evolve quickly in a matter of days. To circumvent this issue I propose a dynamic factor model to estimate both the economic activity latent factor and the effects of anticipated tax shocks on it. The empirical basis for this type of study relies on the joint analysis of text data and high frequency indicators of economic activity. In principle, one may want to consider different frequencies, therefore a mixed frequency analysis is appropriate. To my knowledge this is the first paper that exploits a dynamic factor model to account for fiscal policy effects on economic activity. The factor specification considers the dynamics of the factor and the potential effects of tax changes and their anticipation on it. To identify the effects of anticipations on economic activity I take advantage of the variation of estimated beliefs on future tax changes over time. The variation responds to anticipations of approved tax bills but also to ex-post-erroneous anticipations given the information in the media.

In my empirical analysis, I have control of the potential effects of tax changes from the bill initial status to the implementation. This strategy allows measuring implementation effects of tax changes net of other anticipation mechanisms that could be affecting previous levels of economic activity.

My results reveal that one-month-ahead anticipations of tax approvals significantly stimulate current economic activity growth. A ten percent increment in the measure of one-month ahead tax anticipations reduces the monthly growth rate by 1.5%. Two and three month ahead anticipations revert sign but do not have a statistically significant effect on economic activity. After controlling for mass media anticipation, direct implementation effects of tax changes are reduced in absolute value but still have short-run negative significant effects. The results also suggest an upward bias in the previous estimates of the fiscal multiplier proposed by Romer & Romer (2010) due to the omission of anticipation effects. My results are robust to the inclusion of controls for implementation delays, following the strategy of Mertens & Ravn (2012). I also analyze the effects of tax cuts what stimulates the economy. Anticipating tax increases reduces economic activity by 1,36% while anticipating tax cuts stimulates it by 3,04%.

From Random Forests I learn which text features best predict future tax approvals; I find that those related to the later stages of the bill process are the most relevant covariates for the one-month ahead predictions of tax approvals. The out-of-sample predictive performance of the algorithm reveals we could have predicted four out of twenty tax bill approvals at US Congress by using the television news about taxes. **Relation to the Literature.** This paper contributes to the extensive literature of the effects of fiscal policy on economic activity. In particular, it contributes to the narrative approach and the results from the seminal work of Romer & Romer (2010) by providing an empirical strategy, a methodology and evidence of the effects of fiscal policy on economic activity from initial information spills of a tax change to its implementation.

Mertens & Ravn (2012) study the effects of tax implementation delays by exploiting the variation in the implementation horizon of the *exogenous* tax changes in Romer & Romer (2010). They document that tax increases stimulate economic activity prior to implementation while contractions in economic activity come earlier than if implementation was at the time of approval. In this paper, I also control for implementation delays and still find evidence of anticipation effects. Also Coglianese et al. (2017) provide evidence of anticipation effects of gasoline tax changes in gasoline demand in a framework of policy certainty.

While there is evidence of anticipation effects under certainty of the policy implementation, little is known in terms of behavior under uncertainty. This paper contributes to this literature by considering anticipations of tax changes before bill approvals at Congress, when the media can publish news related to the bill legal process, its economic and political debate. I also distinguish between delayed and on-time implementation of tax changes and find that after considering mass media anticipations of tax approvals, on-time tax changes have no significant effect after implementation in economic activity while delayed tax changes have effects before and after implementation.

This paper contributes to the literature that studies the macroeconomic effects of news content regarding policy announcements by providing a new way of synthesizing many text features of news data into an indicator, in this case of anticipations, and providing a way to estimate its macroeconomic effects. In this literature different works have exploited the number of news mentioning a particular words or dealing with a particular issue to study the macroeconomic effects of news about particular events. Some of these works are Gregori et al. (2016), Bouzgarrou & Chebbi (2015), Caporale et al. (2018), Beetsma et al. (2013) and Gade et al. (2013).

To my knowledge, this is the first application in the literature combining machine learning and forecasting models to study the effects of massive policy-related information from the media on economic activity. Classification algorithms such as Support Vector Machines, Neural Networks or Naïve Bayes have been mostly used for prediction in the field of finance. Manela & Moreira (2017) uses SVM to produce a measures of stock volatility exploiting the information in disaster news. Min & Lee (2005) use it to predict bankruptcy. Arvanitis & Bassiliades (2017) measures investor sentiment using newspaper data and a Naïve Bayes algorithm. Random Forest Classification is a supervised learning algorithm that has been mainly used in fields other than economics such as bioinformatics (Díaz-Uriarte & De Andres, 2006; Yang et al., 2014, June; Khalilia et al., 2011; Saeys et al., 2007; Chen & Liu, 2005) or ecology (Cutler et al., 2007). An application that is closer to economics is the credit scoring study of Van Sang et al. (2016). Recently, there are some economics applications that have exploited Random Forests such as Bajari et al. (2015), Wager & Athey (2017) or Chernozhukov et al. (2018).

This paper also relates to the literature that studies the effects of uncertainty on economic activity. Baker et al. (2016) study the effects of a broad measure of policy uncertainty on economic activity. They offer a specific measure of tax uncertainty constructed with newspapers mentions of economic policy uncertainty keywords. While their aggregate EU index effectively relates to relevant episodes of economic uncertainty, their EPU tax index is not specifically related to the process of tax legislation. It actually contains other sources of uncertainty related to tax policy that are not directly related to the likelihood of tax bills approvals. My measure of tax beliefs differs in that it is meant to capture the likelihood of a tax liability change at a specific date conditional on the current information available, measuring the degree of certainty on a specific policy change. The goal in my empirical strategy is to precisely track the level of predictive information for each potential episode, leaving aside other confounders.

This paper is also related to the literature that studies the effects of media choices on socio economic outcomes. Berg & Zia (2013) and Bursztyn & Cantoni (2016) provide evidence on microeconomic level effects of television content in private consumption and savings and found significant effects. The effects of media news on economic agents have been extensively documented in the field of finance. For example, the works of Tetlock (2007), Tetlock et al. (2008) and Tetlock (2011) have shown how media sentiment and content affects investors or stock market performance.

Methodologically my paper relates to the time series literature that studies the effects of exogenous shocks on macroeconomic indicators. I propose the application of forecasting and nowcasting models to the empirical analysis of policy shocks and macroeconomic news on economic activity. Using these models one can measure the effects of information shocks that naturally happen at monthly, weekly or daily basis on a latent factor of economic activity, helping us to track closely the effects of economic policy. In earlier work, the study of the effects of tax changes on economic activity has been implemented directly using GDP quarterly data. Mixing frequencies not only gives us the possibility of learning about latent economic activity at other frequencies but also to interpret economic activity in terms of GDP. For this reason, I use Mariano & Murasawa (2003) to estimate the effects of taxes and their anticipation in economic activity.

To some extent this paper also relates to the literature that studies the effects of tax salience on taxpayer behavior. Previous literature has found that consumers undereact to taxes that are relatively less salient. Fore example, Chetty et al. (2009) shows that when alcohol tax increases are included in posted prices, alcohol consumption drops more than when the same tax increase is applied at the register. Li et al. (2014) document a similar behavior for gasoline taxes. Baker et al. (2017) provide evidence of tax salience in local newspapers for VAT changes but also on the importance of fine-grained time series data to uncover economic activity effects of tax changes. Finally, the salience of taxes may not only affect the reaction of consumers but also of politicians who may see an opportunity to extend particular taxes which are less salient (Finkelstein, 2009; Cabral & Hoxby, 2012; Goldin & Homonoff, 2013).

Section 2 reviews the literature about fiscal policy effects on economic activity. In Section 3, I explain the institutional framework for empirical study of tax legislation in the US. In Section 4, I explain the data I use. Section 5 presents the empirical approach and the measure of beliefs about tax legislation success using mass media news. In Section 6, I discuss the empirical specification that I use to estimate the effects of taxes on economic activity. Section 7 contains the results and Section 8 concludes.

2 Literature Review

The identification of the causal effects of fiscal policy on economic output is complicated by the fact that the observed variation in fiscal policy is partly driven by the economic cycle. There are two strands in the literature on the measurement of fiscal multipliers, the structural VAR approach and the Narrative approach. Under the structural VAR approach researchers have exploited plausible relations between economic activity indicators and government revenues and spending with the purpose of removing automatic and discretionary components of fiscal changes.

A variety of structural-VAR strategies have been exploited to accomplish this goal. Blanchard & Perotti (2002) stresses the fact that within guarter discretionary responses to output are unlikely and proceed to clean tax and government revenues from automatic responses to output using guarterly data. They make further assumptions on the structure of these relations and find that both increases in taxes and government spending have a negative effect on output. Mountford & Uhlig (2009) contributes with an extension of a sign restriction strategy to identify fiscal shocks. Using the same data as Blanchard & Perotti (2002), they also exploit structural relations between economic and fiscal indicators; government revenues increasing with output a key identifying assumption they use, that is, when both series co-move it must be the case that there has been an improvement in the business cycle. In addition to these sign restrictions, impulse responses of the fiscal variables are assumed to be orthogonal to busyness cycle shocks identified from the co-movements among a list of indicators. They find that negative tax cuts work the best out of possible linear combinations of tax cuts and spending. These authors also consider the identification of anticipation effects of tax and government changes to output by assuming that one year ahead to the implementation of a tax or government change there may be an effect at the time of announcement which is related to the magnitude of the ex-post shock.

The narrative approach identifies tax and government spending shocks from texts containing information about the political and legislative process of the tax change, which can describe the motivation, size and timing of fiscal shocks. The seminal work in this area is Romer & Romer (2010). They identify a list of tax episodes that can be classified as either endogenous or exogenous to the economic cycle. The classification relies on the narrative analysis of several documents from the legislative and executive power while the bill was under elaboration. A tax change is classified as exogenous if it is known not to offset factors pushing away growth from normal. More specifically, tax changes aimed at improving growth in the long run or those made to deal with inherited budget deficits are considered exogenous. To measure the timing and revenue effects Romer & Romer (2010) rely on documents from the Treasury and the Budget of the Government. Sometimes the conference report on the bill is also a good source

for revenue estimates. Ramey (2011) studies the output effects of government spending but she focuses instead on the three large war episodes and the expected public spending related to those episodes as reported in *Business Week*.

None of these approaches are exempt from criticism. A disadvantage of the narrative approach is the fact that estimates of the magnitude of fiscal shocks obtained from gubernatorial documents may be measured with error if the contemporaneous forecast of the fiscal multiplier by the government in charge is itself biased. This is less of a concern for the structural VAR approach as long as one exploits budgetary fiscal revenue data. The best feature of narrative approach is offering a reduction of the ommited variable problem through controlling for many characteristics of fiscal episodes that are contained in publicly available texts. This is a dimension of the empirical study of these effects to which this paper also wants to contribute.

3 Institutional Framework

US federal laws are not passed until they have been discussed, elaborated and approved in different houses of Congress. For the specific case of tax laws, they generally originate as recommendations from the President, which may be announced at the State of the Union Address. However, other House representative may also propose them. If the President proposes a tax change, the Treasury Department will draft it. Usually, this will happen during the first months of the year because the President will only propose one tax bill per year and if not approved within the year it will die, so the process will have to start from scratch. If the proposal comes from another agent, a representative of the House Ways and Means prepares the law proposal or "bill". All tax legislation in the US must originate in the House of Representatives. When the House of Ways and Means receives the proposal it arranges hearings so that people can testify on the proposal. These people are the Secretary of the Treasury, Administrative officials and other groups of interest. The Committee in charge of the bill will meet in executive session after the hearings. In this session they markup the proposal and discuss it openly in public. They write the proposal in legislative language while simultaneously elaborating a plain language report on the motivation that originate the tax bill.

Once the bill and the report are have been completed, the bill is introduced in the House of Representatives for approval. When the tax bill is passed by the House, the Senate Committee will start a similar process to the one of the House of Representatives, but using the bill and report written by the House of Representatives as the starting point. The Senate Finance Committee usually makes amendments that are written into a new report. The amended bill is debated by the entire Senate, which will decide if it is passed in this house. If there are no amendments and the bill is passed, it is sent to the President to be signed and it becomes law. If on the contrary, there the bill has been amended, the modified bill is returned to the House of Representatives, which will accept it or otherwise appoint a Conference Committee to rewrite the differences with the Senate. The new compromised version is voted by the President. If the President vetoes the bill, the House and the Senate may try to override the veto by securing more than two thirds of the favorable votes.

Once new taxes are legislated it may take months to their implementation; that is, to the moment when citizens have the obligation to comply with the new tax legislation. However, there are some special cases where tax legislation can be retroactive, which implies an obligation to satisfy tax changes related to economic activity prior to the law approval. A simplified version of the timing structure of the evolution of a tax bill into tax law is depicted in Figure 1.



Figure 1: Evolution of US Tax Bills

Notes: This diagram outlines the institutional stages of a tax bill in the US from its origination in a proposal stage to its implementation. t states for a particular moment in time. A is the announcement time, which generally is made at the State of the Union Address. H is the moment when the House of Representatives passes the bill in the form of a report. S is the moment the report is passed by the Senate. B is the time when the bill achieves a compromised version and is approved by the House of Representatives and the Senate. 0 is the moment when the bill is approved by the president and it becomes a law. M is the moment of implementation of tax changes, which can be divided in multiple periods.

Throughout this process, there are many opportunities for the general public to access information about the state of the bill and its prospective success. In specific cases, journalists can listen to the live discussion. After each stage, there is also a public release concerning the step that has been passed. The media is the most common channel of information about the state of tax legislation for most people. Some groups of interest may obtain first hand information by attending the hearings or through consultation of official documents. The media may not cover all the legislative process in the same manner neither all the tax episodes. The mass media may tend to select some episodes and stages thereby influencing the type of information that spills to the general public. We expect the mass media to cover episodes that affect a large share of the population, but also episodes which are of more interest to their target audience. In the latter regard, if there is a diversity of mass media we can presume that episodes of relevance to all the sociodemographic spectrum will be covered by some media.

4 Data

To estimate the effects of anticipation of tax episodes on economic activity I combine various datasets. I use Romer & Romer (2010) exogenous tax liability changes to identify exogenous tax episodes in the US between 1945 and 2007. To construct the new measure of anticipations through mass media releases of information I use data from US television from the Vanderbilt Television News Archive. Finally, to estimate the mixed frequency dynamic factor model I use economic activity data.

4.1 Romer & Romer (2010) Exogenous Tax Changes

I use Romer & Romer (2010) (RR) dataset on US tax liability changes which comprises information on the implementation date, magnitude and "exogeneity" of each US tax change taking place within the period 1948 to 2007. By reading Congressional Reports and other gubernatorial documents, RR identify the motivation, timing and magnitude of each approved tax change happening in the US during this period. Using this information they manually classify tax changes into more or less related to the contemporaneous economic activity cycle. An *exogenous* tax liability change is one that is classified as unrelated to the economic cycle.

In Table 1, I present descriptive statistics for the time series of all the RR *exogenous* tax liability changes. There were in total fourty-two *exogenous* tax liability changes

across the period 1968-2007. The average tax liability change was -0.06 per cent of quarterly nominal GDP (QGDP) in the sample period. The standard deviation of tax liability changes is 0.53. I also manually collect data from Romer & Romer (2009) on the approval dates at US Congress of each tax episode in the US. In Table 1, I also provide descriptive statistics for *delayed* tax liability changes, those that were implemented at least one month after the approval date, *on-time* tax liability changes and those implemented the same month of approval. *On-time* tax liability changes where on average -0.03 of QGDP with a standard deviation of 0.47, while *delayed* tax changes where on average -0.07 of QGDP with a standard deviation of 0.55. In Appendix Table 7, I present the same statistic for the period 1945-2007.

Variable	Delayed	Mean	SD	MIN	25th-p	Median	75th-p	MAX	Obs
Tax change	No	-0.03	0.47	-1.11	-0.27	0.16	0.29	0.49	10
Horizon	No	0	0	0	0	0	0	0	10
Tax change	Yes	-0.07	0.55	-1.65	-0.15	0.07	0.26	0.76	32
Horizon	Yes	21.44	20.54	1.00	5.00	15.50	30.50	80.00	32
Tax change	All	-0.06	0.53	-1.65	-0.25	0.08	0.29	0.76	42
Horizon	All	16.33	20.11	0.00	1.00	8.00	24.00	80.00	42

Table 1: Romer & Romer (2010) Exogenous Tax Liability Changes

Notes: Tax change is the estimated magnitude of the *exogenous* tax liability changes measured in dollars by Romer & Romer (2010) divided by the QGDP and expressed in percentage points. Horizon is the number of months between implementation of the tax change and approval of the tax episode. The summary statistics correspond to time series 1968 to 2007.

In Figure 5, I present the time series of all *exogenous* tax liability changes throughout the period 1968 to 2007. Since an episode may contain more than one tax liability change and they are approved in the same date under the same bill, in Table 2 I provide summary statistics for the magnitude of tax episodes in the US. An average episode is -0.15 of QGDP, with 1.14 standard deviation of QGDP. I also offer the time series of episodes by approval date in Figure 6. In Table 8, I provide the month-of-the-year distribution of tax episodes approval and the implementation of tax liability changes, showing tax approvals are spread happened in almost all months but are concentrated in the months of June to August and January. Tax changes are implemented mostly on January and October with a less presence in the rest of the months.

4.2 Television Data

According to *www.classictvhits.com*, in 1960, out of 180 million US population, 47 million households had a TV set. In 1998, out of 276 million US population, 99 million households had a TV set. The first most viewed show in 1960 (*Gunsmoke*) had an estimated audience of 17.5 million households. Interestingly, the total estimated audience for the top 10 most rated shows in 1960 was 138 million households, approximately

	Mean	Std.	Min.	25th-p	Median	75th-p	MAX	Obs
1945-2007	-0.19	0.96	-4.34	-0.37	-0.04	0.32	1.09	36
1968-2007	-0.15	1.14	-4.34	-0.51	0.05	0.57	1.09	20

Table 2: Romer & Romer (2010) Exogenous Episodes

Notes: This table describes the size of Romer & Romer (2010) *exogenous* tax episodes which are defined as the aggregate value of tax liability changes approved under the same tax bill as a share of month of approval GDP.

the same as in 1998. Television reaches a large share of the US population since the mid twentieth century. Finding out whether information funnelled through this channel regarding future tax changes had any impact on economic activity before they even become approved is a question of interest.

I use the Vanderbilt Television News Archive (VTNA) which has collected all evening news shows aired at CBS, NBC, ABC from 1968 to the present and CNN from the 1995 to the present. The evening news shows last for 30 minutes and are broadcast at 6.30pm in the Eastern Zone¹. I collected data on a brief title description, an abstract of text describing the news, the number of seconds it occupied in the screen and its order of appearance for each broadcast piece of news. I collect these data from their website for shows aired from August 1968 (first available data point) to December 2007 (last observation in Romer & Romer (2010) data). These news programs are an interesting

¹Simultaneously broadcast in Central and Eastern Zones and three-hours tape-delayed broadcast in the Pacific Zone. While CNN has single feed that airs in all time zones.

piece of data to study since they are narrowed to 30 minutes everyday, aired to the whole of the US simultaneously and informative of events which are interesting for large section of the US population. Another good aspect of this data is providing data for at least three major television channels through the sample period.

I define the universe of television news relevant to the study of anticipations of tax changes by selecting those pieces of news that contain the stem 'tax' and the surname of any congressman in charge of an *exogenous* Romer & Romer (2010) episode. I use the Congressional Bills Project: 1947-2007 (Adler & Wilkerson (2007)) to find the names of the congressmen in charge of the *exogenous* bills passed at US Congress. Filtering for 'tax' mentions limits and guarantees that the information set contains those news dealing with taxes. The alternative strategy would be to use all the available news published in these media where tax news would be a small fraction that would hardly survive to text processing algorithms. Using the congressmen surnames filter I can produce a measure of tax anticipations restricted to the *exogenous* tax episodes. This strategy helps to the identification of the effects of tax changes and their anticipations on economic activity.

There are on average 5.36 pieces of news per month in the series of tax news. The standard deviation is of 6.48 pieces. The months with tax news had a minimum of 1 and a maximum of 44 pieces. The average piece of tax news has 137.7 words and 754.6 characters. The standard deviation among the pieces of tax news is 80.8 words and 440.6 characters.

In Table 3, I present summary statistics of the salience of tax news in the media measured by total seconds per month. The median space that these news occupy in the TV is 510 seconds per month, but the variation of minutes across months is large. In Figure 7, I depict the monthly time series of seconds, which highlights that there will be differences of coverage across different tax episodes.

	Mean	Std.	Min.	25th-p	Median	75th-p	MAX	Obs
Tax Salience	970.42	1353.30	10	240	510	1100	12630	337

Table 3: VTDA Tax Salience

Notes: Tax salience is measured as the month total of seconds that tax news were broadcast during the period 1968 to 2008. Tax news are defined as those news that mentioned a congressmen in charge of any Romer & Romer (2010) *exogenous* tax bill and the stem 'tax'.

4.3 Economic Activity Data

I use the same economic activity indicators as the ones used in Mariano & Murasawa (2003), namely quarterly real GDP (QGDP) and four monthly coincident indicators, which are detailed in Table 9. I take the first difference of the natural log of each series, multiplied by 100, and construct the growth rates. Table 4 shows descriptive statistics of the growth rate of each series for the period 1968-2007.

	Mean	Std.Dev.	Min.	Max.	
GDP	0.75	0.82	-2.05	3.82	
EMP	0.21	0.71	-3.59	2.38	
IPI	0.15	0.21	-0.77	1.23	
MANU	0.24	0.99	-3.21	3.54	
RPI	0.24	0.54	-3.21	4.00	

Table 4: Descriptive Statistics of Business Cycle Indicators Growth Rates

Notes: This table presents summary statistics for the five economic indicators defined in Table 9 for the period 1968 to 2007. Mean states for the sample average, Std. Dev. for the standard deviation, Min. for the minimum sample value, Max. for the maximum sample value. Economic indicators detailed in Table 9. Source: Federal Reserve Economic Data.

5 A Narrative Measure of Tax Anticipations using Media Data

Next, I turn to present a novel narrative approach to measure the level of anticipation of fiscal changes using media data. Our goal is to find a measure that captures meaning-ful contributions to beliefs about a potential tax bill approval in a future period through the information released in the news. The beliefs may be those of a representative economic agent that watched TV news and saw different tax episode approvals. People may also gather information from sources other to the mass media. An advantage of exploiting the information provided through the TV channel is that it exposes a large share of the population to the same information shocks simultaneously.

There are different dimensions of a tax episode that could be relevant to the anticipated economic response of agents. In this work, I measure on the anticipation of a tax bill approval at Congress because of its first order importance to the study of economic responses to anticipations. However, other dimensions could be explored such as the sign, magnitude, the progressive nature and the different characteristics of the tax base. Tax episodes with different signs may lead to different kinds of anticipation effects, that is, an anticipation of increases in VAT may induce individuals to increase consumption but one of a tax cut may induce individuals to reduce it. I also study the effects of anticipation of episodes of different sign but learning aside other characteristics is left to future work.

The objects of interest are the beliefs about a tax change happening today and some periods ahead conditional on the information today. Let me denote these objects $p_{t+j|t}$ for j = 0, ..., J, where J is the maximum predictable horizon. I define the relevant information set to predict *j*-months ahead tax changes as the collection of pieces of tax news released at a particular month *t*. To estimate the measure of beliefs I exploit the variation of the frequency of relevant text features in the tax news across months. It is key to restrict the focus to *exogenous* tax changes since otherwise fiscal changes may be related to other unobservable variables affecting economic activity, contaminating any empirical results relating tax changes and economic activity.

5.1 Data Processing

To construct measures of beliefs, I created a balanced time series variable of aggregate monthly tax news to which I apply automatic text analysis techniques for preprocessing. The pre-processing stage consists of removing english stopwords², deleting punctuation characters, digits and words of less than three characters and transforming to lower case format all words. Finally, I lemmatize and stem³ the data to reduce the dimension of the vocabulary.

The corpus is the collection of texts of the pre-processed variable, where each text is the collection of stems used at each month. Using the bag-of-words approach, we

²I also remove all names of the congressmen in charge of the tax bills to avoid the problem of overfitting through these text features.

³This first step implies singularizing words and lemmatizing verbs. The second part consists of keeping the stem of each word and removing the rest of information.

represent a month observation *t* by a vector of stem frequencies, $\mathbf{f}_t = (f_1, f_2, ..., f_v, ..., f_V)_t$, where *V* is the maximum size of the vocabulary of the corpus and $f_{v,t}$ is the absolute frequency of stem *v* in month *t*. Hence, we can represent the corpus as $C = {\{\mathbf{f}_t\}_{t=1,...,T}}$ where *T* is the total number of months in the sample. Under this approach, I do not consider word ordering and the matrix *C* is a rather sparse matrix of tokens. *C* is a matrix of *T* Months by *V* features that belong in the corpus.

A final processing step is done into the corpus features to further reduce the dimensionality of the problem; I delete those features that appear in less than 10% of the months in the sample. The purpose of this dimensionality reduction is to eliminate month-specific words that can overfit the data. The size of the resulting dataset is 177 text features with a sparsity of 82%. The time series dimension of the data is of 477 observations.

Figure 2 describes the processed series of tax news by means of a *wordcloud* that represents the relative frequency of all resulting terms in the series. As shown, these news mostly speak about the different stages of the bill process until being signed. Table 10 presents summary statistics for frequency of the top 20 text features in the corpus. The most frequent text feature in the universe of tax news is 'report'. It occurs on average 8.39 times per month with a standard deviation of 18 and is not present in only 263 months. Its maximum monthly frequency is 142 times in a month. It is also very frequent to hear talking about 'tax', it is present in 295 months⁴. The stem for president and other text features representing House or Senate are very frequent. It is also very common to see 'bill' or 'note', 'examin', 'detail','comment', or 'introduc', suggesting that the most common text features in the tax news sample refer to the process of introducing a new tax bill in the US.

This is an instance in which there are more regressors than data points, so that standard econometric models of discrete choice, such as Multinomial Logit or Probit, cannot deal with this problem. To address this quantification challenge I use a machine learning algorithm from the family of Classification and Regression Trees (CART). In particular, I borrow the Random Forest Classifier to predict a tax episode being approved in the future conditional on the information on television tax bill news.

⁴The difference between the number of months with tax news and the number of months with a mention of the tax stem is due to the fact that some tax words such as 'taxpayers' or 'surtax' have stems 'taxpay' and 'surtax', respectively, instead of 'tax'.



Notes: This figure is a *wordcloud* that represents the relative frequency of text features in the cleaned sample of tax news. Larger font size implies more frequency in a ratio 4:0.5. Tax news are defined as those news that mentioned a congressmen in charge of any Romer & Romer (2010) *exogenous* tax bill and the stem 'tax'.

5.2 Random Forest Classifier

The Random Forest (RF) was introduced by Breiman (2001). It is an algorithm that applies bootstrap aggregation to multiple decision trees. A decision tree (in a classification tasks) is a representation for a classifying possibility where there are nodes representing the features we want to use for prediction and branches that can combine the nodes and lead to leafs where there are the classes. This method partitions the feature space and fits a model to each partition. The algorithm has to learn the criteria to split the nodes and pruning the trees; in this way it learns how variables fit the data and which variables are more relevant to fit the data. Two differences distinguish RF from classical decision trees. The first is that within each decision tree of the forest the variables considered to split a node are a random subspace of all the features in the sample. A second important difference is that a RF draws a number of random samples from the training set and estimates a decision tree for each random sample, averaging the results over all the estimated trees. This strategy helps reducing the variance of the estimation and works better in cases where each decision tree has limited bias. Comprehensive discussions on RF are presented in Hastie et al. (2009) and Murphy (2012).

To construct a measure of beliefs on future tax approvals of different signs I estimate a RF where the dependent variable is an indicator variable that has value 1 if at a particular month in the future there was a tax approval at Congress. The independent variables are those in the matrix of text features, *C*. I implement the estimation with the R package 'randomForest' by Liaw & Wiener (2002).

In contrast with a traditional prediction model, such as logit or probit, RF solves the problem of having too many variables. To exploit the text in a traditional model one would have to drastically reduce the dimensionality of the text to a few features according to some arbitrary criteria whereas using RF there is statistical learning on which features better predict among a large list of features, what also improves model fit. It is also attractive in the situation of many predictors because it performs variable selection in a flexible way. RF is one approach among several approaches that have been recently developed to deal with large covariance spaces. For example, Lasso techniques and other penalized flexible regression methods are a popular alternative in econometric application (Belloni et al., 2013). However, Caruana et al. (2008) show that RF classifiers perform better than other models such as SVM, neural nets or boosted trees.

RF were developed in computer science as black-box predictive algorithms. Their properties from a statistical point of view are an active area of research. A recent contribution is the work of Wager & Athey (2018). This paper develops normal asymptotic theory to a random forest model and confidence intervals for random forest predictions. They apply the theory to causal inference of treatment effects with unconfoundedness. As the authors point out, it is the first step in the direction of making random forests tools for statistical inference instead of black-box algorithms. Providing standard errors for the predictions of beliefs produced by RF is out of the scope of this paper.

Text features in the tax news dataset tend to co-occur many times within the series, i.e. the number of months that two different terms are published together is large. For example, words like "new" and "tax" are mentioned together in 291 months, "approval" and "house" in 113 months. This is common across many pairs of words. In addition, some words are likely to be redundant for the prediction of a tax approval and even obscure the prediction power of others if the correlation among them is not properly accounted for. There are two interesting departures from the classical RF algorithm that I explore in this section and enhance model performance to the specific data issues that appear in this paper.

Fuzzy Random Forests I explore the Fuzzy Random Forests (FF) introduced by Conn et al. (2016) in the R implementation 'fuzzyforest'. This approach consists of estimating different classical RF to predict the target variable using different groups of covariates. The groups of covariates can be proposed by the researcher as the outcome of some correlation analysis, such as the proposed Weighted Gene-Coexpression Network Analysis (WGCNA). From each group-level RF the least important variables are discarded by setting a drop parameter. There is a selection RF run using all the screened covariates resulting from the group-level RF. From this last RF the researcher can set another final number of selected covariates to which a final RF is fit. Depending on the number of iterations made into each group-level and the selection RF the computation burden of the algorithm can be large. In the following I show that this algorithm is more suitable for the prediction of tax approvals since it is the case that there are many redundant words in the vocabulary.

5.3 Estimated Beliefs

In this section, I describe the empirical measures of tax anticipations that exploit TV data and Random Forests. In Figures 8 and 9, I present the OOB (out-of-bag) predictions from RF and FF which, in this case, are predicted probabilities of tax episode approvals at t + 1 conditional on the information set at t (computed using the bootstrap samples that the algorithms did not use to estimate the model, so these are out-of-sample predictions of the models). We can note that there are 5 episodes which are predicted by the FF, from the most predicted to the less predicted, we find, the Economic Recovery Act of 1981, the Omnibus Budget Reconciliation Act of 1990, the Omnibus Budget Reconciliation Act of 1993, the Tax Reform Act of 1969 and the Revenue Act of 1971. The RF has lower predictability than the FF because it predicts less episodes. For the rest of the paper we focus in FF results as estimated measures of tax anticipations.

The FF fitted probability of tax approvals at t + 1 conditional on information released at t is presented in Figure 10. Summary Statistics for this measure are presented in Table 5. As we can see, for one-month-ahead anticipation of tax approvals the median belief in the time series is a 0,4%, however, the 95% is 9,7% and the standard deviation is 12.7%. The two-month-ahead and three-month-ahead measures of anticipation have a smaller 95th percentiles and slightly smaller standard deviations. In Figure 11, I

	$P(approval_{t+1} News_t)$	$P(approval_{t+2} News_t)$	$P(approval_{t+3} News_t)$
p25	0	0	0
p50	0.004	0.005	0.005
p75	0.021	0.022	0.021
p95	0.097	0.061	0.070
SD	0.125	0.107	0.114

Table 5: FF Summary Statistics

Notes: This table contains the 25^{th} -percentile (p25), the median (p50), the 75^{th} -percentile (p75), 95^{th} -percentile (p95) of the estimated measures of beliefs for tax approvals at t + 1, t + 2 and t + 3 respectively en each column. SD states for standard deviation.

show which are the text features that have more predictability for one-month-ahead tax anticipations. 'tax', 'presid', 'consid', 'bill' and vocabulary related to the the approval of tax bills at the US Congress are in this top ranking. In Appendix C I expose other results from Fuzzy Forest which might be of interest to the reader.

On the shadow side, the estimated models present relatively poor predictive performance. Using out-of-bag (OOB) samples none of these models predicted the event of a tax change next month with more than 50% probability, however FF was closer at some events. One explanation for this result relates to the limited number of *exogenous* tax episodes to learn from in the sample period, 20 versus the 457 months. Whereas it may seem that a TV viewer watching a news report about tax legislation would likely be able to predict very well a tax approval in the next month, this does not need to be the case because viewers receive different levels of information across episodes.

In Table 11, I provide a comparison of the two models across different measures of goodness of fit. Accuracy and Lift AUC is slightly larger for Classical RF but Gain AUC is larger for FF, however, all the accuracy measures are pretty similar across the two models. However, Log-Loss, MSE, RMSE, PRAUC, RMSE, PRAUC and Zero One Loss are smaller for FF. The two models are pretty similar, slightly higher accuracy for RF while lower loss for FF, what allows us to conclude that FF predictions are better than those of Classical RF.

The anticipatory information contained in the measure of anticipations is a combination of salience of the tax news and the likelihood of approval of the tax bill given the political context. Disentangling between these two channels is challenging given the interconnection between the political decisions and the media. I also estimate measures of tax anticipations for two and three months ahead of potential episode approval. It is also important to note that the current implementations in R were released very recently, so further testing is warranted.

6 A Model for the Effects of Taxes and Anticipations on Economic Activity

In this section, I develop a time series approach to measure the effects of tax changes and their short-term anticipations on economic activity. I start by introducing the general framework and incorporate the different shocks to a dynamic model of economic activity.

6.1 A Model for Monthly Economic Activity

Traditionally, the measure of economic activity used to account for the effects of tax changes has been quarterly GDP. One concern with the use of quarterly GDP is that time aggregation may mask important anticipation effects; after all information disseminated through the media may evolve quickly in a matter of days or weeks. Here, I target a monthly frequency which is economically relevant for the question at hand and empirically feasible, given the indicators of economic activity at my disposal.

Stock & Watson (1991) provided a methodology to construct an index of economic activity exploiting the co-movements of monthly indicators. Later, Mariano & Murasawa (2003) provided a mixed frequency dynamic factor model that allows to exploit quarterly GDP jointly with indicators that have other time frequencies. I borrow their mixed frequency dynamic factor model (MFDFM) which allows me to incorporate data at different frequencies by modeling missing observations corresponding to lower frequency indicators, quarterly GDP in particular. For the purpose of this paper, mixing frequencies has two different advantages. Not only it profits from the quarterly GDP variation to construct the coincident indicator of economic activity but it also recovers an index of economic activity that relates to latent monthly GDP. For all *t*, a one factor model for $y_t^* = (y_{1t}^*, y_{2t})$ is such that:

$$\begin{pmatrix} y_{1,t}^* \\ y_{2,t} \end{pmatrix} = \beta f_t + e_t \tag{1}$$

$$\phi_f(L)f_t = \omega_t \tag{2}$$

$$\phi_e(L)e_t = \epsilon_t \tag{3}$$

$$\begin{pmatrix} \omega_t \\ \epsilon_t \end{pmatrix} \sim NID \begin{pmatrix} 0 & \begin{bmatrix} \sigma_{\omega}^2 & 0 \\ 0 & \Sigma_{\epsilon} \end{bmatrix} \end{pmatrix}$$
(4)

where $\beta \in \mathbb{R}^N$ is a factor loading vector, f_t is a scalar stationary sequence of a common factor, e_t is an N-variate stationary sequence of idiosyncratic shocks or factors, L is a lag operator, $\phi_f(.)$ is a *p*th-order polynomial on \mathbb{R} , $\phi_e(.)$ is a *q*th-order polynomial on \mathbb{R}^{NxN} . The left-hand side variable in the first equation, $y_{1,t}^*$, is monthly GDP which is a latent variable in our model. The second outcome, $y_{2,t}$, is a N-1 vector of observable indicators of economic activity at monthly frequency. σ_{ω}^2 is the variance of the error term in the model for the common factor. Σ_{ϵ} is the variance covariance matrix of the error term in the VAR model for the idiosyncratic shocks. For identification of this model we normalize the first element of β (the one associated with the first economic indicator of the measurement equation) to $\beta_1 = 1$, and we specify Σ_{ϵ} and $\phi_e(.)$ as diagonal matrices. Since $y_{1,t}^*$ is latent, one cannot estimate model (1)-(4), instead, the proposal of Mariano & Murasawa (2003) is to estimate a model with $y_t = (y_{1t}, y_{2t})$, where y_{1t} is quarterly GDP observable every third period and change the specification for the measurement equation of y_{1t} to one that expresses quarterly GDP as the geometric mean of monthly GDP. Hence,

$$y_{1,t} = \beta_1 \left(\frac{1}{3}f_t + \frac{2}{3}f_{t-1} + f_{t-2} + \frac{2}{3}f_{t-3} + \frac{1}{3}f_{t-4}\right) + \frac{1}{3}e_{1,t} + \frac{2}{3}e_{1,t-1} + e_{1,t-2} + \frac{2}{3}3_{1,t-3} + \frac{1}{3}e_{1,t-4}$$

This expression for $y_{1,t}$ comes from defining latent quarterly GDP as the geometric mean of monthly levels, taking logarithms to GDP and expressing quarterly GDP growth in terms of the factor. The result is that our one factor model implicitly constructs a monthly measure of economic activity according to the following aggregation plan:

$$\begin{pmatrix} y_{1,t} \\ y_{2,t} \end{pmatrix} = \begin{pmatrix} \beta_1(\frac{1}{3}f_t + \frac{2}{3}f_{t-1} + f_{t-2} + \frac{2}{3}f_{t-3} + \frac{1}{3}f_{t-4}) \\ \beta_2 f_t \end{pmatrix} + \begin{pmatrix} \frac{1}{3}e_{1,t} + \frac{2}{3}e_{1,t-1} + e_{1,t-2} + \frac{2}{3}3_{1,t-3} + \frac{1}{3}e_{1,t-4} \\ e_{2,t} \end{pmatrix}$$

where the left-hand side will contain only observables and β_2 and e_2 will be vectors of the same dimension as $y_{2,t}$.

6.2 Tax Shocks on Monthly Economic Activity

In this section, I propose a methodology for measuring the effects of tax changes on economic activity at monthly frequency. Romer & Romer (2010) provided evidence on the effects of exogenous tax changes on quarterly GDP using a dynamic linear regression model of quarterly GDP on tax liability changes. Not only the magnitude of the effects and its persistence are relevant features of a policy, but also the time at which the effects come into place. From a policy perspective, one may need to choose between two types of policies so having a methodology that measures the effects at higher frequency than a quarter seems an interesting avenue of work. To estimate the effects of tax liability changes on monthly economic activity I suggest estimating a DFM as (1)-(4) where the specification for the factor process (2) is given by

$$\phi_f(L)f_t = \sum_{s=0}^{S} \theta_s \tau_{t-s} + \tilde{\omega}_t$$
(5)

where τ_{t-s} is a tax liability shock to the economy at time t-s, θ_s accounts for the effects of a tax liability shock happening at time *s* before, *S* is the maximum number of lags to allow for the effects of tax liability changes on GPD. $\tilde{\omega}_t$ is the new error term for the factor. For causal validity it is key to satisfy the identification assumption that $\tilde{\omega}_t$ and τ_{t-s} are independent for all *t*, *s*, and this is the reason why the narrative approach uses RR exogenous tax liability changes instead of all types of tax liability changes during this period.

6.3 Media Anticipation on Monthly Economic Activity

There is considerable amount of information prior to the approval of a tax bill that spills through the mass media and has prediction power on tax bill approvals, as documented in Section 5. Does mass media anticipation have any effects on economic activity prior to the approvals? To answer this question, I measure the effects of the media-based anticipation measure of tax approvals at t + i conditional on the information released at t by specifying the following model for my latent factor of economic activity:

$$\phi_f(L)f_t = \sum_{s=0}^S \theta_s \tau_{t-s} + \sum_{i=0}^I \delta_i p_{t+i|t} + \bar{\omega_t}$$
(6)

where $p_{t+i|t}$ is the media-based anticipation measure for a tax approval happening at t + i conditional on the information at t. δ_i is the effect of unit changes in the beliefs for rises and cuts approvals respectively. This model traces the effects of taxes from initial information releases by the media to the public. It will capture if there is any effect of tax anticipations under uncertainty of the bill being legally approved. Obviously we do not observe the true value of the beliefs, these beliefs are estimated in a previous step as detailed in Section 5. Hence, the factor model with estimated beliefs is

$$\phi_f(L)f_t = \sum_{s=0}^{S} \theta_s \tau_{t-s} + \sum_{i=0}^{I} \delta_i \hat{p}_{t+i|t} + \hat{\omega}_t$$
(7)

where $\hat{p}_{t+i|t}$ is a proxy for the true beliefs on tax legislation approval and $\hat{\omega}_t$ is the new error term. Hence it is crucial that the error term is unrelated not only to implemented tax changes but also mass media beliefs on future tax approvals.

Anticipation of tax approvals associated to tax liability increases are likely to have different effects of those implying tax liability cuts. The joint prediction of sign and approval of a tax episode is challenging given the available data. However, it is not far from reality that people know the sign before the approval of a particular episode. To provide light on the differential effects of anticipations of tax rises with respect to tax cuts I construct an indicator variable that takes the value of 1 when there is a mention of "increase" or "rise" among the tax news. Let denote this indicator s_t .

$$\phi_f(L)f_t = \sum_{s=0}^{S} \theta_s \tau_{t-s} + \sum_{i=0}^{I} \delta_i^- \hat{p}_{t+i|t} + \beta s_t + \sum_{i=0}^{I} \delta_i^+ \hat{p}_{t+i|t} s_t + \dot{\omega}_t$$
(8)

where δ_i^+ and δ_i^- are the effects of anticipated tax approvals conditional on the mention of tax increases in the news or the absence of mention, respectively. β captures the marginal effect of mentioning tax rises in economic activity. $\dot{\omega}_t$ is the new error term.

6.3.1 Controlling for Implementation Delays

In the time between a tax approval and its implementation there is knowledge that a tax liability change is going to take place at a given point in time and this knowledge may affect economic activity. Mertens & Ravn (2012) document that there are significant anticipation effects of tax changes during the months between approval and

implementation of the tax liability changes. Their notion of anticipation refers to the information that makes the economic agent certain of a tax implementation at a given point in time in the future. In contrast, this paper studies anticipations where the agent still has uncertainty about the event of a tax approval at Congress. To avoid confusion here I refer to Mertens & Ravn (2012) classification of tax liability changes according to information as *on-time*, when they refer to changes that are implemented at the same time they are approved at Congress, and *delayed*, when there is one or more periods between the time of approval of the tax change and its implementation.

The omission of this intermediate period of information may contaminate the estimates of the implementation effects of tax changes because previous levels of economic activity may be affected by the knowledge of a future tax change implementation. In addition, this fact potentially may contaminate the effects of media anticipation of taxes if there is some correlation in the data between *delayed* tax changes and the measure of mass media beliefs on a future tax approval. To control for these potential intermediate period effects I follow Mertens & Ravn (2012) strategy to account for the potential effects of *delayed* tax changes versus *on-time* tax changes. In contrast to their work, I use measures of implementation delays computed at monthly frequency. The specification for the factor (2) that controls for all uncertain and certain anticipations on top of the direct implementation effects is,

$$\phi_f(L)f_t = \sum_{s=0}^{S} \theta_s^o \tau_{t-s}^o + \sum_{j=0}^{J} \theta_j^d \tau_{t-j,0}^d + \sum_{m=1}^{M} \lambda_m \tau_{t,m}^d + \sum_{i=0}^{I} \delta_i \hat{p}_{t+i|t} + \dot{\omega}_t$$
(9)

where τ_{t-s}^{o} are *on-time* tax shocks which are implemented at t - s, τ_{t-j}^{d} are *delayed* tax shocks which are implemented at t - j, $\tau_{t,m}^{d}$ are cumulative *delayed* tax shocks at t to be implemented at t + m. *On-time* tax shocks are only part of the information set when implemented, that is at t - s. *Delayed* tax shocks are part of the information set when the laws are approved but they are implemented m periods ahead, thus we track the effects of implementation and the effects of the tax being part of the information set since the law approval using this distinction. Finally $\dot{\omega}_t$ is the new error term. This specification controls for the effects of tax changes since the first spills of information captured by the mass media beliefs until their implementation and posterior dynamics. The identification of these effects is achieved if the error term is uncorrelated to implementation, delayed and anticipation tax shocks. A final specification is the one combining the effects of s_t as in (8) to model (9).

While Romer & Romer (2010) estimate a dynamic regression model for GDP, Mertens & Ravn (2012) estimate a VAR model on quarterly economic activity indicators. Meanwhile, I estimate effects on a latent measure of monthly economic activity that is identified through the MFDFM.

7 Empirical Results

In this section I provide evidence on the effects of exogenous tax changes and their anticipations on economic activity. The estimates are the result from the estimation of the MFDFM detailed at Section 6.1 for various specifications of the factor process (2). To do so I use the Kalman Filter and Maximum Likelihood Estimation. Economic activity series are expressed in first differences of the natural log multiplied by 100, i.e. in growth rates. Table 4 describes the series of economic indicators. Romer & Romer (2010) exogenous tax liability changes are expressed in percentage points of monthly GDP. Table 7 describes tax liability changes in the period 1968-2007 as a whole and disaggregated into unanticipated and anticipated using the definition of Mertens & Ravn (2012) (on-time and delayed in the terminology of this paper). I demean all the series so that the models are estimated without constant terms. I do not standardize the growth rates of the indicators neither the tax and information shocks, so that I can identify the common factor as the latent monthly real GDP growth. Before estimation, I follow Mariano & Murasawa (2003) and substitute missing observations of the every-third-period observable y_{1t} with random draws from a standard normal distribution.

7.1 Implementation Effects of Tax Changes

Figure 3 presents the implementation effects of *exogenous* tax liability changes on economic activity, in the period 1948 to 2007, following model (5) for the factor. I control for up to S = 36 period dynamics, that is, for three year dynamics as in Romer & Romer (2010). The figure shows the cumulative effects in terms of an increase in tax liabilities of a one percent of QGDP together with the one-standard-error bands. The maximum effects are achieved 29 months after implementation of the tax changes when monthly economic activity growth drops by 99.56%. Given that average monthly GDP growth

was approximately 0.29% in 1948-2007, the maximum implementation effect of a 1% of QGDP increase of tax liabilities is a reduction of monthly economic activity growth of 0,28%. There are also immediate effects of the tax changes on monthly GDP growth. For example, two months after implementation tax increases reduce monthly economic activity growth by 15.8%. The monthly dynamics and magnitude of the effects are comparable to the baseline results of Romer & Romer (2010), which are expressed on a quarterly basis and with respect to the level of GDP.



Figure 3: Tax Changes on Economic Activity

Notes: This figure shows the cumulative effect of a one percent percent increase in tax liabilities over monthly GDP of the Romer & Romer (2010) *exogenous* tax changes.

For the period that we dispose of television data, 1968 to 2007, immediate implementation effects are -6.6% for monthly economic activity growth. Two months after implementation the effects are -10.7%. Maximum effects are a -69.1% and happen 25 months after implementation.

7.2 Media Anticipation on Economic Activity

This section presents evidence on the effects of mass media anticipations of tax bills approvals on economic activity. The effects are captured by the parameter δ_j for j = 1, 2, 3 of model (7); that is, the effects of a marginal change in the probability of a

tax approval happening one to three months ahead on current economic activity conditional on all the relevant information of tax news in the media. Baseline results use the estimated measure of anticipations from the FF algorithm. Mass media induced beliefs about a tax approval next month significantly affect current economic activity as documented in Table 6. I quantify that a ten percent probability of a tax approval at the following month increases the growth rate by 1.5%. Beliefs on tax approvals happening two or three periods ahead revert sign and are poorly significant.

	(1)	(2)	(3)	(4)
P(Tax Approval at t+1)	0.155	0.1698	0.304	0.303
	(0.099)	(0.097)	(0.125)	(0.122)
P(Tax Approval at t+2)	-0.039	-0.0008	-0.050	-0.025
	(0.115)	(0.115)	(0.156)	(0.155)
P(Tax Approval at t+3)	-0.026	-0.0314	-0.091	-0.101
	(0.109)	(0.108)	(0.117)	(0.116)
Tax Increase at t			0.003	-0.009
			(0.034)	(0.035)
P(Tax Approval at t+1)*Tax Increase at t			-0.443	-0.407
			(0.209)	(0.210)
P(Tax Approval at t+2)*Tax Increase at t			-0.019	0.025
			(0.234)	(0.234)
P(Tax Approval at t+3)*Tax Increase at t			0.486	0.523
			(0.306)	(0.298)
Observations	471	471	471	471

Table 6: Media Anticipation Effects on Current Economic Activity

Notes: This table contains the effects of one to three month ahead media anticipation of tax approvals on current economic activity using the estimated beliefs of a FF. Column (1) presents the results from the factor model affected by RR exogenous tax implementations plus beliefs on tax bills approvals. Column (2) presents the results when the factor is differentially affected by *on-time* and *delayed* tax liability changes and implementation delays on top of media anticipation. Columns (3) and (4) present of model () and () but it distinguishes the effects of anticipations of net tax cuts from net tax rises.

Figure 4 presents the cumulative implementation effects of a one percent increase in tax liabilities after controlling for mass media anticipation of tax bill approvals up to three months ahead, together with one-standard-error bands. There are some quantitative differences with respect to the results presented of Section 7.1. The one-month implementation effects become -6.6% of monthly EA, two months after implementation tax changes reduce monthly EA by 10.35% and the maximum implementation effect is -65.32% of montly EA, smaller in absolute value to the maximum effect of tax changes if one does not consider media anticipations. However, maximum effects are also reached 25 months after implementation. After considering the measures of tax anticipations, the direct implementation effects of tax changes are reduced in absolute value suggesting an upward bias in previous estimates.



Figure 4: Tax Changes on Economic Activity controlling for Media Anticipations

Notes: This figure shows the cumulative effect of a one percent percent increase in tax liabilities after controlling for the new measure of media anticipations of the Romer & Romer (2010) *exogenous* tax episodes.

As documented by Mertens & Ravn (2012), in some tax episodes, there is a lapse of time between the approval and the implementation of tax changes, which I define as implementation delay. To the purpose of this paper, I use the term anticipations only for predictions of taxes under uncertainty while these authors refer to anticipations once it is certain the future implementation date. I estimate the model for M = 18, as in Mertens & Ravn (2012). In column (2) of Table 6 I also present the results of estimating MFDFM with specification (9) for the factor. I distinguish between delayed and on-time tax liability changes and include tax implementation delays. The absolute value of the effects of media anticipation slightly increase in this specification, so not controlling for the Mertens & Ravn (2012) does not confound the effects of media anticipations. In Figures 12 and 13, I show that implementation effects of on-time tax changes are not significantly different from zero and the they are reduced in absolute value. The effects of delayed tax changes are significant are larger than the original implementation effects of tax changes of Figure 4. Up to seven month implementation delays of tax rises have a positive and significant impact on economic activity of 40% of monthly EA. Longer horizon implementation delays are poorly estimated as shown in the Figure.

7.3 Heterogeneous Effects of Tax Anticipations

In this section, I provide evidence of the differential effects for anticipation of tax cuts and tax rises. Conditional on the media release of information about a potential tax approval, it is likely that people is aware of what is the net tax liability change associated to the potential approval since media also makes reference to terms like "increase", "rise" or "cut". There are 20 episode approvals in the sample and learning how to predict the sign joint to the approval based on 10 approvals per sign resulted in something unfeasible. I construct an indicator variable that captures the mention of "increase" or "rise" within the tax news to approximate the possibility of a tax rise approval. The average of this variable is 0.23 and the standard deviation is 0.42, it takes the value 1 in 10% of the months approximately.

In columns (3) and (4) I control for this indicator and its interaction with media anticipation of tax approvals for the classical Romer & Romer (2010) and the Mertens & Ravn (2012), respectively. A 10% probability of tax approvals conditional on the tax news at t not mentioning tax increases significantly stimulates current monthly economic activity growth by 3.04%. In the case of the media mentioning tax increases the effect is a reduction of monthly economic activity growth by 1.36%. In column (4) the effects are 3.03% and 1.12% respectively what implies that controlling for implementation delays slightly reduces the magnitude of the effects of anticipation of tax increases, however the effects are still statistically significantly different from zero. The implementation effects of tax changes for the results of column (3) are -6.14%, -10.19% and -67.11% for one-month, two-month and maximum effects respectively, hence they are similar to those corresponding to the results of column (1).

8 Conclusions

This paper introduces a new empirical measure that captures the level of anticipation of tax bill approvals in the US for the period 1968-2007. I combine text data and machine learning techniques to construct a measure that enables the study of anticipation effects of tax changes on economic activity following the steps of the narrative approach.

Since time aggregation in the analysis of economic indicators may mask important anticipation effects and information may evolve quickly in a matter of days, the paper proposes a mixed frequency dynamic factor model to estimate both the economic activity latent factor and the effects of anticipated tax shocks on it. To my knowledge this is the first paper that exploits a dynamic factor model to account for fiscal policy effects on economic activity.

This work constributes to the study of the macroeconomic effects of available information of different policies prior to their legal approval by providing a strategy that exploits information in the news and combines it with other lower frequency economic indicators using a well-estabilished methodology.

My results reveal that one-month-ahead anticipations of tax approvals significantly stimulate current economic activity. A ten percent increment in the measure of onemonth ahead anticipations reduces the monthly growth rate by 1.5%. Two and three month ahead anticipations revert sign but do not have a statistically significant effect on economic activity. After controlling for mass media anticipation, direct implementation effects of tax changes are reduced in absolute value but still have short-run negative significant effects. I also analyze the effects of the anticipation of tax increases versus tax cuts finding that it is the anticipation of tax cuts what stimulates the economy.

Media coverage of tax episodes at particular dates may be related to unobservable factors that relate to economic activity. I overcome this concern by aggregating the news in a month and assuming that tax announcements can be postponed but not replaced within the month. Finally, there may have been anticipation of eventually not-approved tax bills. In principle, my measure of beliefs can capture them as long as congressmen in charge of those where also in charge of some approved bill. Capturing the effects for this sort of episodes may be worth as a piece of evidence of unconventional fiscal policy but I do not dispose of data that allows me to identify those specific episodes.

The measure of tax anticipations captures both information about tax salience and that of the likelihood of approval of a tax bill at Congress. The strong relation between both channels of information challenges the study of their separate effects, postponing this question to future work. Finally, Random Forests, as other machine learning algorithms, were developed in computer science as black-box predictive algorithms and their properties from a statistical point of view are currently an active area of research. Providing standard errors on the output of Random Forests is something I hope could be addressed in future research.

Appendix A Additional Figures



Figure 5: Exogenous Tax Liability Changes

Notes: This figure shows the time series of Romer & Romer (2010) *exogenous* tax liability changes as a share of quarterly GDP (percentage points) at implementation months. The time series starts at January 1945 and ends at December 2008.

Figure 6: Tax Episodes



Notes: This figure shows the time series of Romer & Romer (2010) *exogenous* tax episodes as a share of quarterly GDP (percentage points) at approval months. The time series starts at January 1945 and ends at December 2008.





Notes: This figure shows the time series of television salience of tax news expressed in seconds using the VTDA sample of news. The time series starts at August 1968 and ends at December 2008.





Notes: This figure shows the OOB predicted probability of media anticipations for tax bill approvals at Congress at t + 1 conditional on VDTA Tax News at t for the period July 1968 to December 2007 and using standard Random Forests.





Notes: This figure shows the OOB predicted probability of media anticipations for tax bill approvals at Congress at t + 1 conditional on VDTA Tax TV News at t for the period July 1968 to December 2007 using Fuzzy Forests.





Notes: This figure shows the fitted predicted probability measure of media anticipations for tax bill approvals at Congress at t + 1 conditional on VDTA Tax TV News at t for the period July 1968 to December 2007 using Fuzzy Forests.



Figure 11: Feature Importance in the one-month-ahead Fuzzy Forest

Notes: This figure describes the ranking of feature importance for classification of one-period-ahead months as months of tax approvals in US Congress conditional on VDTA Tax TV News at current month. The right-hand-side figure uses Mean Decrease Gini while the left-hand-side figure uses Mean Decrease Accuracy to measure feature importance for prediction. From the top to the bottom, more to less relevant features.

Figure 12: On-time Tax Changes on Economic Activity



Notes: This figure shows the cumulative effect of a one percent percent increase in on-time tax liabilities of Romer & Romer (2010) *exogenous* tax changes, distinguishing between on-time and delayed tax changes as in Mertens & Ravn (2012) and also controlling for media tax anticipations.



Figure 13: Delayed Tax Changes on Economic Activity

Notes: This figure shows the cumulative effect of a one percent percent increase in delayed tax liabilities of Romer & Romer (2010) *exogenous* tax changes, distinguishing between on-time and delayed tax changes as in Mertens & Ravn (2012) and also controlling for implementation delays (as in Mertens & Ravn (2012)) and media tax anticipations.

Appendix B Additional Tables

Variable	Delayed	Mean	SD	MIN	25th-p	Median	75th-p	MAX	Obs
Tax change	No	-0.23	0.56	-1.83	-0.34	-0.19	0.12	0.49	21
Horizon	No	0	0	0	0	0	0	0	21
Tax change	Yes	-0.04	0.52	-1.65	-0.15	0.09	0.31	0.76	39
Horizon	Yes	20.72	19.33	1.00	5.00	16.00	29.00	80.00	39
Tax change	All	-0.11	0.54	-1.83	-0.26	0.05	0.24	0.76	60
Horizon	All	13.47	18.44	0.00	0.00	5.00	20.50	80.00	60

Table 7: Romer & Romer (2010) Exogenous Tax Liability Changes

Notes: Tax change is the estimated magnitude of the *exogenous* tax liability changes measured in dollars by Romer & Romer (2010) divided by the QGDP and expressed in percentage points. Horizon is the number of months between implementation of the tax change and approval of the tax episode. The summary statistics correspond to time series 1945 to 2007.

Month of Year	Approval Month	Implementation Month
1	16.67	66.67
2	5.56	0
3	0	3.33
4	5.56	3.33
5	2.78	0
6	11.11	5
7	16.67	8.33
8	13.89	5
9	8.33	0
10	8.33	6.67
11	8.33	1.67
12	2.78	0
Events	36	60

Table 8: Monthly Tax Activity

Notes: This table contains the relative frequency for tax approvals (1) and tax implementations (2) in particular months of the year across the sample period 1945-2007.

Indicator	Description
000	GDP quarterly, seasonally adjusted annual rate, deflated with
GDP	Implicit Price Deflator of GDP (Index 2009=100, quarterly, seasonally adjusted)
EMP	All employees: total nonfarm payrolls, thousands of persons,
	monthly, seasonally adjusted
IDI	Industrial Production Index, Index 2012=100, monthly,
IFI	seasonally adjusted
ΜΑΝΗΙ	Real Manufacturing and Trade Industries Sales, millions of
MANU	chained 2009 Dollars, monthly, seasonally Adjusted
RPI	Real personal income excluding current transfer receipts, billions
	of chained 2009 Dollars, monthly, seasonally adjusted annual rate

Table 9: US Business Cycle Indicators

Notes: Definition of economic activity indicators used in this paper. Source: FRED, Federal Reserve Bank of St. Louis.

_

	Mean	Std. Dev.	Median	Max	Not null
report	8.39	18	1	142	263
tax	6.61	12.79	2	98	295
presid	4.45	10.29	1	92	240
say	4.18	8.06	1	61	257
senat	2.71	6.53	0	62	190
repres	2.65	7.87	0	78	189
note	2.32	5.94	0	52	187
hous	2.23	6.26	0	70	180
comment	1.86	6.66	0	76	153
introduc	1.79	5.34	0	62	191
congr	1.75	4.14	0	38	161
plan	1.75	4.72	0	46	145
cut	1.68	5.5	0	62	137
budget	1.68	9.01	0	164	109
bill	1.58	5.41	0	67	121
show	1.55	4.6	0	55	182
democrat	1.48	4.62	0	51	129
give	1.43	3.09	0	29	193
state	1.38	3.76	0	54	175
examin	1.24	3.56	0	46	157

Table 10: Top 20 Text Features

Notes: This table describes the absolute monthly frequency of the twenty most frequent text features int the corpus of tax news for the period 1968 to 2008. There are 489 (month) observations in the sample period. Tax news are defined as those news that mentioned a congressmen in charge of any Romer & Romer (2010) *exogenous* tax bill and the stem 'tax'. Mean states for the sample average, Std. Dev. for the standard deviation, Median for the median, Sum for the total sample occurrence, Range for the difference between the minimum and maximum value in the sample, Not null for the total months of no occurrence, Obs. for the month observations

	Classical RF	Fuzzy Forests
Accuracy	0.958	0.957
AUC	2.067	2.067
Gain AUC	0.507	0.508
Lift AUC	0.714	0.691
KS	91.525	91.525
Log-Loss	0.676	0.358
MSE	0.040	0.039
PRAUC	0.678	0.655
RMSE	0.200	0.199
RMSLE	0.139	0.139
Zero One Loss	0.852	0.686

Notes: This table contains a list of measures of goodness of fit for Random Forests Classifiers comparing the results for OOB Classical RF, Test WSRF, OOB FF results. Accuracy is the share of true positive and true negative predictions made over the total of test samples. AUC is the area under the ROC curve, AUC is the area under the curve and measures the probability that a classifier will rank a randomly chosen positive instance higher than a randomly chosen negative one, Gain AUC, Lift AUC measures how much more likely a positive responses is received than a randomly chosen response, KS is score/probability band where separate between positives and negatives is maximum. Log-loss is the negative log-likelihood of the true labels given a probabilistic classifiers predictions. MSE is the mean square error of predicted probability of classes versus true classes. PRAUC is the area under the precision-recall curve, RMSE is the root mean squared error of the predicted probability of classes versus true classes, RMSLE is the root mean squared logarithmic loss. Zero One Loss is the normalized classification error loss.

Appendix C Fuzzy Forests Results

In this section I explain the details of the estimation of Fuzzy Forests (FF) to predict a tax approval at the US Congress using media data.

In a first stage, this algorithm suggests the estimation of Weighted Gene-Coexpression Network Analysis to the original set of covariates to identify groups of correlated features. This first stage can be rephrased as estimating an average hierarchical clustering model using as input data a matrix of topological overlap dissimilarities, which is a way of representing the networks of the original data set. Setting a threshold level, the algorithm finds a number of modules (or groups) of variables that would relate more closely. I decide to do WGCNA on the text features alone, since I do not want the algorithm to disregard variables such as past salience or episodes. I chose power of 3 and minimum module size of 20 to avoid having too many groups. The resulting modules are described in Figures 14 to 17 using word clouds that represent the relative frequency of features in the corpus for each modules.

The turquoise module contains words mostly related to 'report', 'hous', 'presid', 'senate', which are instituional features of the process of the bill approval at US Congress. The brown module represents words related mostly to 'introduc', 'comment', 'campaign', 'mention', that is, words that deal with different comments at the initial debate of a bill. The blue module is a group of words related to other issues that interact to tax changes in the process of debate. Finally, the grey module is a small group about some government departments and past vocabulary.

For each module, a Recursive Feature Elimination Random Forests (RFE-RF) is estimated to screened out the least predictive features. The researcher has to decide how many features are eliminated at each iteration and how many features wants to keep. I set to elimitate 1% of the features at each iteration until having dropped 75% of the original number of features in the module. After the screening step, the algorithm runs a final RFE-RF using all the screened features from the different modules to predict the dependent variable. In this stage, the interaction between the different features is taken care of. Here, I also eliminate a 1% of screened features until I keep the best 50 features according to the algorithm. Figure 18 presents the relative importance before and after the screening of features. The most present module is the turquoise followed by the blue and the brown. The red part of the bars is the amount of words finally selected from each module after feature selection. The grey module

almost dissapears after feature selection. This selection of features is consistent with the predictability of the features since the turquoise group is more relate to the event of approval of a bill while the brown module relates to the initial process of bill approval where there is considerable uncertainty about the potential approval. Finally, I estimate a classical RF using the 50 selected features.

Figure 14: Blue Word Cloud

secretari would way busin problem chairperson policilast may rate cost_{inflat} unitcontrol pay o job new Ū think ne C hixon σ economi money committe

Figure 15: Brown Word Cloud

comment mention introduc price democrat elect talk econom governor cite issu republican leader e polit scene nation californium candid examin feder Show



Figure 17: Turquoise Word Cloud







References

- Adler, E. S., & Wilkerson, J. (2007). Congressional bills project: 1947–2000. NFS 00880066 & 00880061.
- Arvanitis, K., & Bassiliades, N. (2017). Real-time investors? sentiment analysis from newspaper articles. In *Advances in combining intelligent methods* (pp. 1–23). Springer.
- Bajari, P., Nekipelov, D., Ryan, S. P., & Yang, M. (2015). Demand estimation with machine learning and model combination (Tech. Rep.). National Bureau of Economic Research.
- Baker, S. R., Bloom, N., & Davis, S. J. (2016). Measuring economic policy uncertainty. *The Quarterly Journal of Economics*, *131*(4), 1593–1636.
- Baker, S. R., Johnson, S., & Kueng, L. (2017). *Shopping for lower sales tax rates* (Tech. Rep.). National Bureau of Economic Research.
- Beetsma, R., Giuliodori, M., de Jong, F., & Widijanto, D. (2013). Spread the news: The impact of news on the european sovereign bond markets during the crisis. *Journal of International Money and Finance*, *34*, 83 101.
- Belloni, A., Chernozhukov, V., & Hansen, C. B. (2013). Inference for high-dimensional sparse econometric models. In *Advances in economics and econometrics: Tenth world congress* (Vol. 3, p. 245-295). Cambridge University Press.
- Berg, G., & Zia, B. (2013). Harnessing emotional connections to improve financial decisions: Evaluating the impact of financial education in mainstream media. *World Bank Policy Research Working Paper*(6407).
- Blanchard, O., & Perotti, R. (2002). An empirical characterization of the dynamic effects of changes in government spending and taxes on output. *The Quarterly Journal of Economics*, *117*(4), 1329-1368.
- Bouzgarrou, H., & Chebbi, T. (2015). Does news on the euro area impact the sovereign yield spreads? *International Journal of Monetary Economics and Finance*, *8*(1), 4–19.

Breiman, L. (2001). Random forests. *Machine learning*, 45(1), 5–32.

- Bursztyn, L., & Cantoni, D. (2016). A tear in the iron curtain: The impact of western television on consumption behavior. *Review of Economics and Statistics*, *98*(1), 25–41.
- Cabral, M., & Hoxby, C. (2012). *The hated property tax: salience, tax rates, and tax revolts* (Tech. Rep.). National Bureau of Economic Research.
- Caporale, G. M., Spagnolo, F., & Spagnolo, N. (2018). Macro news and bond yield spreads in the euro area. *The European Journal of Finance*, *24*(2), 114–134.
- Caruana, R., Karampatziakis, N., & Yessenalina, A. (2008). An empirical evaluation of supervised learning in high dimensions. In *Proceedings of the 25th international conference on machine learning* (pp. 96–103).
- Chen, X.-W., & Liu, M. (2005). Prediction of protein–protein interactions using random decision forest framework. *Bioinformatics*, *21*(24), 4394–4400.
- Chernozhukov, V., Demirer, M., Duflo, E., & Fernandez-Val, I. (2018). *Generic machine learning inference on heterogenous treatment effects in randomized experiments* (Tech. Rep.). National Bureau of Economic Research.
- Chetty, R., Looney, A., & Kroft, K. (2009). Salience and taxation: Theory and evidence. *The American economic review*, *99*(4), 1145–1177.
- Coglianese, J., Davis, L. W., Kilian, L., & Stock, J. H. (2017). Anticipation, tax avoidance, and the price elasticity of gasoline demand. *Journal of Applied Econometrics*, *32*(1), 1–15.
- Conn, D., Ngun, T., & Ramirez, C. M. (2016). Fuzzy forests: Extending random forests for correlated, high-dimensional data. *Journal of Statistical Software*.
- Cutler, D. R., Edwards, T. C., Beard, K. H., Cutler, A., Hess, K. T., Gibson, J., & Lawler, J. J. (2007). Random forests for classification in ecology. *Ecology*, *88*(11), 2783–2792.
- DAcunto, F., Hoang, D., & Weber, M. (2016, August). *The effect of unconventional fiscal policy on consumption expenditure* (Working Paper No. 22563). National Bureau of Economic Research.

- Díaz-Uriarte, R., & De Andres, S. A. (2006). Gene selection and classification of microarray data using random forest. *BMC bioinformatics*, *7*(1), 3.
- Finkelstein, A. (2009). E-ztax: Tax salience and tax rates. *The Quarterly Journal of Economics*, *124*(3), 969–1010.
- Gade, T. P., Salines, M., Glöckler, G., & Strodthoff, S. (2013, July). 'loose lips sinking markets?': The impact of political communication on sovereign bond spreads (ECB Occasional Paper No. 150). European Central Bank.
- Goldin, J., & Homonoff, T. (2013). Smoke gets in your eyes: cigarette tax salience and regressivity. *American Economic Journal: Economic Policy*, *5*(1), 302–336.
- Gregori, W. D., Sacchi, A., et al. (2016). Has the grexit news spilled over into euro area financial markets? the role of domestic political leaders, supranational executives and institutions (Tech. Rep.). Money and Finance Research group (Mo. Fi. R.)-Univ. Politecnica Marche-Dept. Economic and Social Sciences.
- Hastie, T. J., Tibshirani, R. J., & Friedman, J. H. (2009). *The elements of statistical learning : data mining, inference, and prediction.* New York: Springer.
- Khalilia, M., Chakraborty, S., & Popescu, M. (2011). Predicting disease risks from highly imbalanced data using random forest. *BMC medical informatics and decision making*, *11*(1), 51.
- Li, S., Linn, J., & Muehlegger, E. (2014). Gasoline taxes and consumer behavior. *American Economic Journal: Economic Policy*, *6*(4), 302–342.
- Liaw, A., & Wiener, M. (2002). Classification and regression by randomforest. *R News*, *2*(3), 18-22.
- Manela, A., & Moreira, A. (2017). News implied volatility and disaster concerns. *Journal of Financial Economics*, *123*(1), 137–162.
- Mariano, R. S., & Murasawa, Y. (2003). A new coincident index of business cycles based on monthly and quarterly series. *Journal of Applied Econometrics*, 18(4), 427–443.

- Mertens, K., & Ravn, M. O. (2012). Empirical evidence on the aggregate effects of anticipated and unanticipated us tax policy shocks. *American Economic Journal: Economic Policy*, 4(2), 145-81.
- Min, J. H., & Lee, Y.-C. (2005). Bankruptcy prediction using support vector machine with optimal choice of kernel function parameters. *Expert systems with applications*, 28(4), 603–614.
- Mountford, A., & Uhlig, H. (2009). What are the effects of fiscal policy shocks? *Journal* of Applied Econometrics, 24(6), 960–992.
- Murphy, K. P. (2012). Machine learning: a probabilistic perspective. MIT press.
- Ramey, V. A. (2011). Identifying government spending shocks: It's all in the timing*. *The Quarterly Journal of Economics*, *126*(1), 1-50.
- Romer, C. D., & Romer, D. H. (2009). A narrative analysis of postwar tax changes. Unpublished paper, University of California, Berkeley (June).
- Romer, C. D., & Romer, D. H. (2010). The macroeconomic effects of tax changes: Estimates based on a new measure of fiscal shocks. *American Economic Review*, *100*(3), 763-801.
- Romer, C. D., & Romer, D. H. (2016). Transfer payments and the macroeconomy: The effects of social security benefit increases, 1952-1991. *American Economic Journal: Macroeconomics*, 8(4), 1-42.
- Saeys, Y., Inza, I., & Larrañaga, P. (2007). A review of feature selection techniques in bioinformatics. *bioinformatics*, 23(19), 2507–2517.
- Stock, J., & Watson, M. W. (1991). A probability model of the coincident economic indicators. In *The leading economic indicators: New approaches and forecasting records* (p. 63-90). Cambridge University Press.
- Tetlock, P. C. (2007). Giving content to investor sentiment: The role of media in the stock market. *The Journal of Finance*, *62*(3), 1139–1168.

- Tetlock, P. C. (2011). All the news that's fit to reprint: Do investors react to stale information? *Review of Financial Studies*, *24*(5), 1481–1512.
- Tetlock, P. C., Saar-Tsechansky, M., & Macskassy, S. (2008). More than words: Quantifying language to measure firms' fundamentals. *The Journal of Finance*, *63*(3), 1437–1467.
- Van Sang, H., Nam, N. H., & Nhan, N. D. (2016). A novel credit scoring prediction model based on feature selection approach and parallel random forest. *Indian Journal of Science and Technology*, 9(20).
- Wager, S., & Athey, S. (2017). Estimation and inference of heterogeneous treatment effects using random forests. *Journal of the American Statistical Association*(just-accepted).
- Wager, S., & Athey, S. (2018). Estimation and inference of heterogeneous treatment effects using random forests. *Journal of the American Statistical Association*, *0*(0), 1-15.
- Yang, J., Yao, D., Zhan, X., & Zhan, X. (2014, June). Predicting disease risks using feature selection based on random forest and support vector machine. In *Bioinformatics research and applications: 10th international symposium* (pp. 1–11). Springer International Publishing.

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 1720 LUIS J. ÁLVAREZ and ANA GÓMEZ-LOSCOS: A menu on output gap estimation methods
- 1721 PAULA GIL, FRANCISCO MARTÍ, JAVIER J. PÉREZ, ROBERTO RAMOS and RICHARD MORRIS: The output effects of tax changes: narrative evidence from Spain.
- 1722 RICARDO GIMENO and ALFREDO IBÁÑEZ: The eurozone (expected) inflation: an option's eyes view.
- 1723 MIGUEL ANTÓN, SERGIO MAYORDOMO and MARÍA RODRÍGUEZ-MORENO: Dealing with dealers: sovereign CDS comovements.
- 1724 JOSÉ MANUEL MONTERO: Pricing decisions under financial frictions: evidence from the WDN survey.
- 1725 MARIO ALLOZA: The impact of taxes on income mobility.
- 1726 DANILO LEIVA-LEON: Measuring business cycles intra-synchronization in US: a regime-switching interdependence framework.
- 1727 PIERRE GUÉRIN and DANILO LEIVA-LEON: Model averaging in Markov-Switching models: predicting national recessions with regional data.
- 1728 MÁXIMO CAMACHO and DANILO LEIVA-LEON: The propagation of industrial business cycles.
- 1729 JAMES COSTAIN: Costly decisions and sequential bargaining.
- 1730 MARIO ALLOZA: Is fiscal policy more effective in uncertain times or during recessions?
- 1731 PIERRE GUÉRIN and DANILO LEIVA-LEON: Monetary policy, stock market and sectoral comovement.
- 1732 HENRIK JENSEN, IVAN PETRELLA, SØREN HOVE RAVN and EMILIANO SANTORO: Leverage and deepening business cycle skewness.
- 1733 CÉSAR MARTÍN MACHUCA: External stress early warning indicators.
- 1734 RODOLFO G. CAMPOS: International migration pressures in the long run.
- 1735 ANDREA ARIU, ELENA BIEWEN, SVEN BLANK, GUILLAUME GAULIER, MARÍA JESÚS GONZÁLEZ, PHILIPP MEINEN, DANIEL MIRZA, CÉSAR MARTÍN MACHUCA and PATRY TELLO: Firm heterogeneity and aggregate business services exports: micro evidence from Belgium, France, Germany and Spain.
- 1736 LEONARDO GAMBACORTA, STEFANO SCHIAFFI and ADRIAN VAN RIXTEL: Changing business models in international bank funding.
- 1737 ENRIQUE MORAL-BENITO and FRANCESCA VIANI: An anatomy of the Spanish current account adjustment: the role of permanent and transitory factors.
- 1738 MARÍA J. NIETO and LARRY D. WALL: Cross-border banking on the two sides of the Atlantic: does it have an impact on bank crisis management?
- 1739 JACOPO TIMINI: Currency unions and heterogeneous trade effects: the case of the Latin Monetary Union.
- 1740 PAULINO FONT, MARIO IZQUIERDO and SERGIO PUENTE: Subsidising mature age employment or throwing coins into a wishing well: a quasi-experimental analysis.
- 1741 THOMAS FUJIWARA and CARLOS SANZ: Norms in bargaining: evidence from government formation in Spain.
- 1742 ENRIQUE ALBEROLA, ÁNGEL ESTRADA and FRANCESCA VIANI: Global imbalances from a stock perspective.
- 1743 ÓSCAR ARCE, RICARDO GIMENO and SERGIO MAYORDOMO: Making room for the needy: the credit-reallocation effects of the ECB's corporate QE.
- 1744 M. D. GADEA-RIVAS, ANA GÓMEZ-LOSCOS and EDUARDO BANDRÉS: Clustering regional business cycles.
- 1745 NEZIH GUNER, YULIYA KULIKOVA and JOAN LLULL: Marriage and health: selection, protection, and assortative mating.
- 1746 SERGIO MAYORDOMO and MARÍA RODRÍGUEZ-MORENO: Did the bank capital relief induced by the supporting factor enhance SME lending?
- 1747 KATALIN BODNÁR, LUDMILA FADEJEVA, MARCO HOEBERICHTS, MARIO IZQUIERDO PEINADO, CHRISTOPHE JADEAU and ELIANA VIVIANO: Credit shocks and the European Labour market.
- 1748 PIERRE GUÉRIN, DANILO LEIVA-LEON and MASSIMILIANO MARCELLINO: Markov-switching three-pass regression filter.
- 1749 ISABEL ARGIMÓN: Decentralized multinational banks and risk taking: the Spanish experience in the crisis.
- 1750 BING XU: Permissible collateral and access to finance: evidence from a quasi-natural experiment.
- 1751 GERGELY AKOS GANICS: Optimal density forecast combinations.
- 1801 OLYMPIA BOVER, LAURA HOSPIDO and ERNESTO VILLANUEVA: The impact of high school financial education on financial knowledge and choices: evidence from a randomized trial in Spain.
- 1802 IGNACIO HERNANDO, IRENE PABLOS, DANIEL SANTABÁRBARA and JAVIER VALLÉS: Private Saving. New Cross-Country Evidence Based on Bayesian Techniques.

- 1803 PABLO AGUILAR and JESÚS VÁZQUEZ: Term structure and real-time learning.
- 1804 MORITZ A. ROTH: International co-movements in recessions.
- 1805 ANGELA ABBATE and DOMINIK THALER: Monetary policy and the asset risk-taking channel.
- 1806 PABLO MARTÍN-ACEÑA: Money in Spain. New historical statistics. 1830-1998.
- 1807 GUILHERME BANDEIRA: Fiscal transfers in a monetary union with sovereign risk.
- 1808 MIGUEL GARCÍA-POSADA GÓMEZ: Credit constraints, firm investment and growth: evidence from survey data.
- 1809 LAURA ALFARO, MANUEL GARCÍA-SANTANA and ENRIQUE MORAL-BENITO: On the direct and indirect real effects of credit supply shocks.
- 1810 ROBERTO RAMOS and CARLOS SANZ: Backing the incumbent in difficult times: the electoral impact of wildfires.
- 1811 GABRIEL JIMÉNEZ, ENRIQUE MORAL-BENITO and RAQUEL VEGAS: Bank lending standards over the cycle: the role of firms' productivity and credit risk.
- 1812 JUAN S. MORA-SANGUINETTI and ROK SPRUK: Industry vs services: do enforcement institutions matter for specialization patterns? Disaggregated evidence from Spain.
- 1813 JAMES CLOYNE, CLODOMIRO FERREIRA and PAOLO SURICO: Monetary policy when households have debt: new evidence on the transmission mechanism.
- 1814 DMITRI KIRPICHEV and ENRIQUE MORAL-BENITO: The costs of trade protectionism: evidence from Spanish firms and non-tariff measures.
- 1815 ISABEL ARGIMÓN, CLEMENS BONNER, RICARDO CORREA, PATTY DUIJM, JON FROST, JAKOB DE HAAN, LEO DE HAAN and VIKTORS STEBUNOVS: Financial institutions' business models and the global transmission of monetary policy.
- 1816 JOSE ASTURIAS, MANUEL GARCÍA-SANTANA and ROBERTO RAMOS: Competition and the welfare gains from transportation infrastructure: evidence from the Golden Quadrilateral of India.
- 1817 SANDRA GARCÍA-URIBE: Multidimensional media slant: complementarities in news reporting by US newspapers.
- 1818 PILAR CUADRADO, AITOR LACUESTA, MARÍA DE LOS LLANOS MATEA and F. JAVIER PALENCIA-GONZÁLEZ: Price strategies of independent and branded dealers in retail gas market. The case of a contract reform in Spain.
- 1819 ALBERTO FUERTES, RICARDO GIMENO and JOSÉ MANUEL MARQUÉS: Extraction of inflation expectations from financial instruments in Latin America.
- 1820 MARIO ALLOZA, PABLO BURRIEL and JAVIER J. PÉREZ: Fiscal policies in the euro area: revisiting the size of spillovers.
- 1821 MARTA MARTÍNEZ-MATUTE and ALBERTO URTASUN: Uncertainty, firm heterogeneity and labour adjustments. Evidence from European countries.
- 1822 GABRIELE FIORENTINI, ALESSANDRO GALESI, GABRIEL PÉREZ-QUIRÓS and ENRIQUE SENTANA: The rise and fall of the natural interest rate.
- 1823 ALBERTO MARTÍN, ENRIQUE MORAL-BENITO and TOM SCHMITZ: The financial transmission of housing bubbles: evidence from Spain.
- 1824 DOMINIK THALER: Sovereign default, domestic banks and exclusion from international capital markets.
- 1825 JORGE E. GALÁN and JAVIER MENCÍA: Empirical assessment of alternative structural methods for identifying cyclical systemic risk in Europe.
- 1826 ROBERTO BLANCO and NOELIA JIMÉNEZ: Credit allocation along the business cycle: evidence from the latest boom bust credit cycle in Spain.
- 1827 ISABEL ARGIMÓN: The relevance of currency-denomination for the cross-border effects of monetary policy.
- 1828 SANDRA GARCÍA-URIBE: The effects of tax changes on economic activity: a narrative approach to frequent anticipations.

BANCODE ESPAÑA

Eurosistema

Unidad de Servicios Auxiliares Alcalá, 48 - 28014 Madrid E-mail: publicaciones@bde.es www.bde.es